Author’s response to reviews

Title: Comparison of The French and CARDS Classifications for Lumbar Degenerative Spondylolisthesis: Reliability and Validity

Authors:

Chao Kong (353397342@qq.com)
Xiangyao Sun (sun.xiang.yao@163.com)
Junzhe Ding (18310693576@qq.com)
Machao Guo (guomachao23@gmail.com)
Xiangyu Li (lxyyxl@sina.com)
Shibao Lu (spinelu@163.com)

Version: 4 Date: 04 Aug 2019

Author’s response to reviews:

Dear editor,

Thank you very much for your suggestions. The questions are responded as following:

1. Thank you very much. We have made the changes. The corresponding email is spinelu@163.com.

2. The missing information has been added in the authors contributions.

3. The method part has been revised to minimize overlap.

4. All tables were cited in the manuscript (labled in red).

5. We have stated “Not Applicable” in this section.

6. We have removed all tracked changes.

7. We have removed the “revision note” and inserted a “figure legend” heading.

Thank you again for your time and patience!

Lu Shibao